Author's response to reviews

Title: Will Mobile Diabetes Education Teams (MDETs) in Primary care Improve Patient Care Processes and Health Outcomes?: Study Protocol for a Randomized Controlled Trial

Authors:

Enza Gucciardi (egucciari@ryerson.ca)
Mariella Fortugno (mariella.fortugno@ryerson.ca)
Stacey Horodezny (shorodezny@thc.on.ca)
Wendy Lou (wendy.lou@utoronto.ca)
Souraya Sidani (ssidani@ryerson.ca)
Sherry Espin (sespin@ryerson.ca)
Fiona Webster (fiona.webster@utoronto.ca)
Baiju R Shah (baiju.shah@ices.on.ca)

Version: 2 Date: 5 September 2012

Author's response to reviews:

The revisions made in this version of the manuscript (September 5th, 2012) are as follows:

- The title has been revised to follow the format: ___________: study protocol for a randomized controlled trial.

- The name of the ethical approval body and a statement that we will gain informed consent from all participants has been included in our methods section.

- Tables have been removed from the body of the manuscript and submitted as additional files.

- The order of authors' names has been revised.